

Received 6 October 2023, accepted 16 November 2023, date of publication 28 November 2023, date of current version 6 December 2023.

Digital Object Identifier 10.1109/ACCESS.2023.3336761



Semi-Passive UHF RFID Sensor Tags: A Comprehensive Review

HECTOR SOLAR^{®1}, ANDONI BERIAIN^{®1}, (Member, IEEE), ROC BERENGUER^{®1}, (Senior Member, IEEE), JAVIER SOSA^{®2}, AND JUAN A. MONTIEL-NELSON^{®2}, (Senior Member, IEEE)

¹Department of Electrical and Electronic Engineering, School of Engineering, University of Navarra (Tecnun), 20018 San Sebastian, Spain ²Institute for Applied Microelectronics (IUMA), University of Las Palmas de Gran Canaria, 35017 Las Palmas de Gran Canaria, Spain

Corresponding author: Hector Solar (hsolar@unav.es)

This work was supported by project MOONLIGHT through the Spanish Ministry of Science and Innovation under Grant PID2020-117251RB-C21/C22.

ABSTRACT This paper presents a comprehensive overview and analysis of the state-of-the-art (SoA) in semi-passive or Battery-Assisted (BAP) Ultra-High Frequency (UHF) Radio Frequency Identification (RFID) sensor tags compliant with EPC Global G2/ISO-18000C. These tags operate on the same communication principle as fully passive sensor tags but incorporate a battery or an energy harvesting module. This additional power source extends communication ranges and enables power demanding applications using low-power microcontrollers (MCUs) and higher-end sensors. This article also analyzes various key features, including tag integrated circuit (IC) architecture, types of energy harvesting modules, and communication range. The main conclusions are threefold. Firstly, selecting the appropriate tag IC requires a careful analysis of its features such as sensitivity, sensor interfaces, or data logging capabilities. For instance, among the solutions examined in the SoA, half of them opted for a tag IC capable of MCU communication via SPI or I2C buses. Secondly, it is essential to assess both the forward and backward communication links to leverage the sensitivity of the tag IC in BAP mode. Interestingly, only one-third of the SoA solutions achieved the theoretical communication range anticipated by the sensitivity of the tag IC. Finally, an energy budget analysis is required to ensure that the energy generation suffices to meet the energy requirements of the tag. While most solutions rely on batteries as the energy source and analyze battery lifespan, only a few studies employing energy harvesters conduct an energy budget analysis due to the additional complexity involved.

INDEX TERMS Radio frequency identification, battery-assisted sensor tag, energy harvesting, Internet of Things.

I. INTRODUCTION

The UHF RFID technology has gained significant popularity in the Internet-of-Things (IoT) industry due to its cost-effectiveness, extended lifespan, and recent sensing capabilities [1]. This technology has found applications in various fields including logistics, healthcare, transport, and industry, among others. Nevertheless, despite its widespread adoption, a major drawback of the UHF RFID technology lies

The associate editor coordinating the review of this manuscript and approving it for publication was Maurice J. Khabbaz^(D).

in its limited system communication range, which is further exacerbated when new sensors are integrated into the system.

As shown in Fig. 1a, fully passive UHF sensor tags typically comprise an antenna, matching network, tag IC and sensors. Specifically, the tag IC is responsible for communicating with the reader through backscatter modulation of the incoming RF waveform transmitted by the reader. Furthermore, the tag IC also harvests energy from the same incoming RF waveform to power both itself and the sensors by means of an output voltage supply pin (VREG in Fig. 1a), eliminating the need for batteries. Consequently, passive tags exhibit unlimited lifespan, compact size, and low cost. However, due



FIGURE 1. UHF RFID Sensor tag architectures in fully passive (a) and semi-passive (b) configurations.

to the limited energy harvested, fully passive RFID sensor tags have a short communication range of only a few meters, as the energy is shared between the tag IC and the sensors. Moreover, the little energy harvested restricts the type of sensors that can be used.

As shown in Fig. 1b, semi-passive or BAP UHF sensor tags incorporate an external power source. Consequently, semipassive UHF sensing technology overcomes the limitations of fully passive solutions because it enables the utilization of a wider range of sensors trough a low-power MCU interface, without compromising the communication range and still ensuring long autonomy. In addition, recent developments have explored the adoption of energy harvesting modules instead of batteries, potentially offering unlimited autonomy. Moreover, improvements in tag IC designs have introduced a BAP mode by adding an input voltage pin (VBAT in Fig. 1b). This pin externally supplies the internal blocks of the tag IC, thereby providing enhanced sensitivities and consequently improving the communication range of the system. Fig. 2 shows a photo of a SoA example of a semi-passive UHF RFID sensor tag. It implements the architecture depicted in Fig. 1b with a tag IC, an MSP430 low-power MCU and a TMP112 I2C-based digital temperature sensor. It also includes a printed dipole antenna and holders for coin batteries [2].

This work then focuses on the growing UHF semi-passive or BAP sensor tag solutions in the SoA that incorporate the widely adopted EPC C1G2/ISO 18000-6C standard. The paper examines the intended applications, architectures, sensing capabilities and communication ranges of these solutions. Additionally, it provides an in-depth analysis of the performances and features of the available tag ICs. Lastly, the type and performance of the energy source, whether a battery or an energy harvesting module, are also evaluated.

The paper is organized as follows. Section II provides an overview of semi-passive sensor tags within the SoA. Section III presents a detailed analysis of the architecture and performance of the tag ICs found in the SoA. Section IV examines the performance of the energy sources of the sensor tags, whether they are battery-assisted or employ an energy harvester module. Section V addresses the architectures and performance of the sensor tag with special focus on its communication range. Finally, Section VI concludes this paper.

II. SoA OF SEMI-PASSIVE SENSOR TAGS

Table 1 presents the key features of the EPC Global G2/ISO-18000C UHF semi-passive sensor tags in recent years. Table 1 is organized based on the tag IC manufacturer and publication year. As observed, except for a few custom solutions, most of the solutions use commercial off-the-shelf

IEEE Access



FIGURE 2. Example of a UHF RFID semi-passive sensor tag.

TABLE 1. SoA of Semi-Passive Sensor TAGS.

Ref Year	IC Supplier	IC Model	Energy Source	Sensing Parameter	Sensor Type	Sensor Interface	Application
[3] 2016	Impinj	Monza X 8K	Battery	Accelerometer	Digital	MCU-SPI	Infrastructure Monitoring
[4] 2016	Impinj	Monza X 8K	RF/PV	Temperature/Humidity	Digital	MCU-I2C	General
[5] 2019	Impinj	Monza X 8K	Battery	Accelerometer/Strain	Digital/Analog	MCU-SPI/ADC	Infrastructure Monitoring
[6] 2013	Impinj	Monza X 2K	RF	Temperature	Analog	MCU-ADC	General
[7] 2014	Impinj	Monza X 2K	RF/Battery	Accelerometer/Light/Temperature	Digital/Analog	MCU-I2C	General
[8] 2014	Impinj	Monza X 2K	Battery	Accelerometer/Light/Temperature	Digital/Analog	MCU-I2C	General
[9] 2017	Impinj	Monza X	Battery/PV	Temperature	Analog	Analog (Z-mismatch)	Supply Chain
[10] 2014	AMS	SL900A	Battery	Temperature/Pressure	Built-in/Digital	MCU-SPI	General
[11] 2019	AMS	SL900A	Battery	Temperature/Sweat	Analog	Built-in ADC	Healthcare
[12] 2019	AMS	SL900A	RF	Temperature	Built-in	-	General
[13] 2019	AMS	SL900A	PV	-	-	-	General
[14] 2022	AMS	SL900A	RF	Temperature	Analog	Built-in ADC	Healthcare
[15] 2011	EM Micro.	EM4325	Battery	Moisture	Analog	MCU-ADC	Packaging Surveillance
[16] 2019	EM Micro.	EM4325	PV	Temperature	Built-in	-	Buildings
[17] 2020	EM Micro.	EM4325	PV	Temperature	Built-in	-	General
[18] 2021	EM Micro.	EM4325	Battery	IMU	Digital	MCU-SPI	Healthcare
[19] 2021	EM Micro.	EM4325	Battery	Temperature/Heart Rate	Digital	MCU-I2C	Healthcare
[20] 2021	Farsens	Rocky100	PV	Light Emitting Diode	Analog	Analog (Built-in VREG)	Industry
[21] 2022	Farsens	Rocky100	TEG	Temperature	Digital	MCU-I2C	Industry
[22] 2022	Farsens	Rocky100	RF	Temperature/Humidity	Digital	MCU-SPI	Agriculture
[23] 2017	Farsens	ANDY100	TEG	Temperature	Analog	MCU-ADC	Healthcare
[24] 2017	NXP	G2iL	Battery	Strain	Analog (R-to- freq Conv.)	Analog (DC/DC Conv.)	Infrastructure Monitoring
[25] 2015	Fujitsu	MB97R804B	Battery	Accelerometer/Light/Temperature	Digital	MCU-I2C	Transport
[26] 2010	-	Custom	Battery	Temperature	Built-in	-	General
[27] 2013	-	Custom	Battery	ECG	Analog	Built-in ADC	Healthcare
[28] 2015	-	Custom	Battery	Temperature/Humidity	Digital	MCU-I2C	General
[29] 2016	-	Custom	Battery	ECG/Others	Analog	Built-in ADC	General

tag ICs. It is also interesting to see that the list of tag IC manufacturers is short. As shown, the Impinj MONZA 8K/2K is the most widely used, followed by the AMS SL900A, the EM Microelectronic EM4325, and the Farsens Rocky 100. Other solutions are also found with NXP and Fujitsu ICs. The reasons behind the selection of these ICs are discussed in detail in the next section.

Table 1 further provides information on the sensing parameters, sensor types and their interfaces with the tag. A wide range of sensed parameters is observed, including temperature, humidity, acceleration, and others, utilizing both analog and digital sensors. In addition, since the available power is not limited only to the one harvested from the incoming RF waveform combined sensing parameters can be implemented as in [7], [8], and [25], in which acceleration, light and temperature sensor are monitored in the same tag. For the same reason, it is interesting to see that most SoA solutions instead of directly connecting the sensor to the tag IC, they



FIGURE 3. Tag IC Internal block diagram.

include an MCU in between for greater design flexibility. It is worth mentioning that temperature sensing predominates. The main reason, along with the fact that temperature is a basic parameter, is that temperature sensors can be implemented on standard CMOS processes with very low power consumption and therefore integrated with the tag IC. Several ICs like the [33] or the [34] have internal sensor, so temperature monitoring is straightforward in this case.

Table 1 also provides information of the different energy sources employed. Given its voltage stability and long-term energy availability, the majority solutions include a battery. However, to provide a semi-passive batteryless solution, several works incorporate energy harvesting modules with a variety of energy sources: Photovoltaic (PV), Radio Frequency (RF) and Thermoelectric Generators (TEG).

Finally, Table 1 shows that the solutions in the SoA addresses many different applications: industry, transport, healthcare, monitoring of buildings, goods, or infrastructure, etc.

III. TAG IC: ARCHITECTURE AND PERFORMANCE

As mentioned in section II, the list of commercially available UHF tag ICs used in semi-passive sensor tags is not long. The main reason for this is that these ICs fall under the EPC standard's class-3 category, which denotes battery-assisted (BAP) or semi-passive solutions. These ICs are specifically designed to handle and store sensor data and use additional power sources [30]. Fig. 3 illustrates the block diagram of a class 3 semi-passive tag IC. Similar to passive UHF RFID tag ICs, class 3 semi-passive tag ICs consists of two main blocks: the analog front-end for energy harvesting and data communication, and the digital core that implements the EPC C1G2/ISO 18000-6C standard. Within the analog front-end, the Voltage Multiplier (VM) along with the Voltage Limiter transform the incoming RF waveform into a limited DC signal, subsequently charging the C_{SUPPLY} capacitor. The Power Management Unit (PMU) monitors de V_{SUPPLY} level and enables the Voltage Regulators for supplying both the digital core and the active blocks of the analog front-end: ASK Demodulator, PSK Modulator, Clock Generator, Bandgap Reference, etc. All these blocks are also present in passive UHF tag ICs but the main difference in this case is that the V_{SUPPLY} node is externally accessible through a PSM (Power Supply Management) block, so that an external source can be used to power the tag IC. This improves the sensitivity of the tag IC and allows additional features to the IC, such as a digital interface for sensors or data logging. Next a detailed description of these features is presented:

A. BAP SENSITIVITY

This characteristic provides semi-passive tag ICs with a crucial advantage over passive devices. It means that the tag IC presents two modes: passive mode and BAP mode, with improved sensitivity in the latter.

In passive mode, the read sensitivity of the tag IC depends on the voltage multiplier to generate enough DC voltage (V_{SUPPLY} in Fig. 3) at its output considering its load conditions. The load conditions refer to the amount of current required to power the internal active tag IC blocks (ASK Demodulator, PSK Modulator, Clock Generator, Voltage Regulators, Bandgap Reference, etc.) and the external devices (MCU and sensors).

In this passive mode, the current needed to power the active internal blocks of the tag IC comes from transmitted signal of the reader. Without any external load, the sensitivity of the tag IC corresponds to the value shown in Table 2, it is typically

	-								
Ref	Manuf.	Model	Passive Sensitivity (dBm)	BAP Sensitivity (dBm)	Sensor Interface	Battery Supply (V)	Memory (bits)	Current Rdy State (uA)	Functionality
[31]	Impinj	Monza X 8K	-19.1	-26.1	I2C Slave	1.6–3.6	8192	15	RF events /Private data/Dual RF port/ Data protection
[32]	Impinj	Monza X 2K	-17	-24	I2C Slave	1.6–3.6	2176	15	RF events /Private data/Dual RF port/ Data protection
[33]	AMS	SL900A	-7	-15	SPI Slave/ Analog	1.5-3	8416	50	Built-in Temperature Sensor/ Data Logger and RTC//Data protection/ADC
[34]	EM Micro.	EM4325	-8.3	-17/-31	SPI Slave/ Master	1.25-3.65	8192	6	Built-in Temperature Sensor/ Data Logger and RTC/RF events/ Data protection
[35]	Farsens	Rocky100	-14	-24/-35	SPI Slave/ Master	1.4-3	1008	7	LDO with VDD Monitor/Data Logger and RTC/Tamper Alarm/ RF events/Data protection
[36]	Farsens	ANDY100	-4	-	SPI Master	-	128	15	LDO
[37]	NXP	G2iL	-17.6	-27	-	1.8	-	7	Tamper Alarm/Data protection
[38]	Fujitsu	MB97R804B	-6	-	SPI Slave	2.3-3.6	3424	70	Data protection

TABLE 2. Tag IC features.

obtained by querying only the tag's ID, so only the current consumption of the tag IC is considered. However, when an external load is added to the IC, the sensitivity deteriorates. An example of this fact in passive mode is found in [20], whose sensitivity degrades from -14 dBm to -9 dBm when the tag IC is loaded with a LED. Also, in [28] the sensitivity is reduced from -15.1 dBm to -11.2 dBm when getting data samples from the internal temperature sensor if compared to a simple tag ID reading operation. Finally, in [35] the sensitivity is reduced from -14 dBm to -10 dBm when a load of 5 μ A at 1.8 V is used. In semi-passive sensor tags, even though the tag IC is in passive mode, there is a read range improvement when compared to a fully passive sensor tag. As the battery or the energy harvesting module supplies power to the MCU and the sensors, it alleviates the load conditions of the tag IC only to the power required for the tag IC itself. Thus, the sensitivity of the tag IC is not degraded, preserving the read range.

In tag ICs operating in BAP mode, the external supply is not only applied to the MCU and sensors but also to the internal blocks of the IC. The consequence is that there is now no need to wait for the voltage multiplier to provide enough DC output voltage, as the DC voltage is already available from the battery or the energy harvester. Specifically, since the ASK demodulator is externally powered, it can adequately demodulate the information carried in low power signals. Therefore, sensitivity is improved, because those low power signals that could not energize the tag IC now, in BAP mode, can be demodulated correctly because the tag IC is already powered. This is the main reason of the sensitivity enhancement in the BAP mode.

Table 2 summarizes the performance of the tag ICs found within the SoA of semi-passive applications. As observed,

the majority of these tag ICs can be classified as class 3, as they feature a sensitivity improvement from passive to BAP mode. As shown in Table 2, both [31] and [32] report 7 dB sensitivity improvement from -19.1 dBm to -26.1 dBm and from -17 dBm to -24 dBm, respectively. Similarly, in [33] an improvement of 7 dB from -7 dBm to -15 dBm is achieved. A slightly larger improvement is shown in [37] from -17.6 dBm to -27 dBm. Interestingly, [34] and [35] offer a range of sensitivity values in BAP mode. In [34], the BAP sensitivity can be chosen from four different levels (-17 dBm, -22 dBm, -28 dBm or -31 dBm) by means a control word. In [35], it is possible to choose between a sensitivity of -17 dBm for the BAP mode or an enhanced BAP sensitivity of -35 dBm. Exceptions are [36] and [38], which lack BAP mode and thus do not provide enhanced sensitivity. Nevertheless, these two tag ICs are used in the literature because, as explained before, even in passive mode, the addition of an external power source improves the reading range by relaxing the load conditions. Finally, in [38] although there is a supply voltage pin, it is not intended for BAP mode but for memory access, which is FRAM type.

B. EXTERNAL SUPPLY

An essential feature of a tag IC for semi-passive application is an input for external supply, enabling the utilization of an external battery or an energy harvester module. Furthermore, it also desirable that the input pin for external supply may accept a wide range of input voltage levels to accommodate different types of power sources, an eventual voltage reduction in battery voltage or potential voltage fluctuations in energy harvesters. As observed in Table 2 most of the tag ICs fulfills this requirement. The ICs in [31] and [32] allow 1.6 V to 3.6 V voltage range. Similar ranges are found in [33] and [35], spanning from 1.5 V to 3 V and from 1.4 V to 3 V, respectively. The widest range is provided by [34], from 1.25 V to 3.65V. Again, there are some exceptions such as [37], which lacks an input voltage range because it only accepts 1.8V. As aforementioned, the voltage range in [38] is only to allow the FRAM memory access. Finally, [36] has no external supply.

C. SENSOR INTERFACE

As shown in Table 2, since these ICs are designed for sensor data communication, they commonly incorporate I2C (Inter-Integrated Circuit), as in [31] and [32], and SPI (Serial Peripheral Interface) as in [33], [34], [35], [36], and [38], as standard communication buses with sensors. Furthermore, most of these ICs implement Slave modules [31], [32], [33], and [38]. This indicates the intention to communicate with an MCU that usually acts as the Master. The reason for incorporating an MCU is that due to the extra energy source, most solutions do not connect the sensor directly to the tag IC but through a low power MCU because it adds extra flexibility to the types of sensors that can be implemented. By using a low power MCU, multiple I2C, SPI communication ports are available for connecting both the tag IC and digital sensors. Furthermore, the MCU usually includes Analog-to-digital converters (ADCs) for interfacing with analog sensors. Consequently, almost any off-the-self digital or analog sensor can be interfaced. Moreover, the I2C or SPI buses support multiple slave devices, allowing for the addition of new sensors with minimum changes. Finally, the internal memory of the MCU can be used as an additional data logger with greater memory size and lifespan than the one provided by the tag IC.

However, it is worth noting that some ICs, such as [34] [35], incorporate an SPI bus that can be also configured as Master for direct communication with digital sensors, which normally act as Slave devices, so it avoids the MCU. This would eliminate the need for an intermediate MCU if a direct connection with the sensors is desired.

D. USER DATA LOGGING

Many of the ICs allocate internal memory for user data logging purposes. In general, memory depth is not extensive, but it allows the storage in a non-volatile memory (NVM). Additionally, if an RTC is also available, data can be connected to time stamps, enhancing the tag's capability for data, event, or alarm logging. For example, it can be used for event-triggered data logging from an external device such a sensor or a switch. As indicated in Table 2, most of the tag ICs allocate a portion of the internal memory for user data storage. The memory size ranges from 128 bit to over 8 Kbit. The ICs in [31], [33], and [34] shows the highest memory size, whereas [32], [35], and [38] offers memory sizes in the Kbit range. It is worth noting that [37] lacks memory space

for user data, but improved models of the same manufacturer allow 640 bit for this purpose [39].

Finally, it is important to note that the number of erase/write cycles in the NVM are limited to 10^4 or 10^5 cycles, thus imposing a constraint on the lifespan of the sensor tag, particularly when a relatively high data storing rate is anticipated [8]. In that case, using an intermediate MCU is preferred.

E. ADDITIONAL FEATURES

Additionally, several ICs provide additional functionalities, including:

- Internal LDO: This feature ensures a stable voltage supply to an external device, eliminating the need for an external LDO (Low Dropout) regulator. This feature is shown in [35] and [36]. Additionally, [35] incorporates a voltage monitor that activates the LDO regulator only when a sufficient supply voltage is available.
- Analog-to-Digital Converter: An integrated Analogto-Digital Converter enables the digitization of analog sensors. In this case, only [33] includes an ADC.
- Anti-tampering: To safeguard against unauthorized external modifications of data stored in the tag's memory. The ICs in [35] and [37] incorporate anti-tampering measures.
- **RF event:** This functionality enables the tag IC to trigger an event in response to a reader command. For instance, it can be utilized to wake up an external device, such as an MCU, from sleep mode. Solutions in [31], [32], and [35] incorporate this feature.
- Extra RF port: Some ICs like [31] and [32] include an additional RF port, which effectively eliminates orientation-related missed reads or blind spots by utilizing omnidirectional antennas.
- Data protection. This feature allows for the configuration of writing protection on the RFID tag. It offers options such as open writability, password protection, or permanent locking. The majority of ICs allow data protection [31], [32], [33], [34], [35], [37], and [38].
- Internal temperature sensor: Some ICs have an internal temperature sensor as in [33] and [34]. In addition to its potential use as an environmental sensor, it also serves the purpose of tag IC self-protection by monitoring temperature levels.

In summary, the final application of the semi-passive sensor tag, along with the types of sensors, data logging requirements, read range or the energy source employed are the characteristics to consider for proper tag IC choice.

IV. ENERGY SOURCE: ARCHITECTURE AND PERFORMANCE

This section discusses the architectures and performance of the energy sources employed in the SoA. The discussion starts with the solutions that use a battery, followed by the works that employs ambient energy harvesters.

TABLE 3. Semi-passive sensor tags with battery.

Ref.	Avg. Power cons. (μW)	Batt. Voltage (V)	Batt Capacity (mAh)	Lifetime (years)
[5]	-	3	650	-
[8]	20.8	3	225	3.7
[8]	13.7	3	225	5.6
[10]	-	3.7	200	-
[15]	2.1	3	33	5.4
[15]	2.1	3	10	1.7
[18]	-	3	225	2
[24]	-	3.3	750	0.001
[26]	0.36	2.4	1	0.75
[27]	12	1.4	605	5.7
[29]	1	1.8	10	2

A. BATTERY AS THE ENERGY SOURCE

Table 1 presented above also provides an overview of the energy sources employed in the semi-passive solutions. As it has been mentioned, batteries are the most common since it provides stable energy and high autonomy given the low power consumption expected in semi-passive sensor tags [3], [5], [7], [8], [9], [10], [11], [15], [18], [19], [24], [25], [26], [27], [28], [29]. Now Table 3 summarizes the works that offer details about the battery capacity and output voltage, system power consumption and system lifetime.

As it is shown. battery capacities as low as a typical coin cell of 225 mAh or less can expand the sensor tag lifetime for several years, depending on the system duty cycle which impacts on the average power consumption. Examples are given in [8], where the lifetime ranges from 3.7 years for an average power consumption of 20.8 μ W and a sensor data sampling rate of 2 samples/min, to 5.6 years for an average power consumption of 13.7 μ W and a sensor data sampling rate of 1 sample/min. This last result is similar to [27] with a lifetime of 5.7 years for an average power consumption 12.1 μ W and 1.6% duty cycle for a 605 mAh battery capacity.

In [15], the lifetime spans to 5.4 years for a small flexible battery of 33 mAh and 1.7 years for a 10 mAh flexible battery. The sampling rate is of 2 samples/h in this case, which explains the low average power consumption. Works in [26] and [29] are similar in that they both report long battery life for small batteries. Work in [26] reports the smallest capacity of 1 mAh for 9 months of battery life. It must be noted that this calculation is made considering only the tag IC standby current, which explains such low average power consumption. Still, this solution includes a battery recharging circuit to collect the excess of energy from the incoming RF signal. In [29] a battery life of 2 years for a 10 mAh capacity is reported. The low average power consumption is explained by an expected duty cycle of only 0.1%. However, the energy budget is comprehensive in this case because this work also considers the leakage current of the tag IC and, as is the previous case, the contribution of the excess of the incoming RF signal for battery recharging.

The work in [24] shows a different case as it reports a relatively short lifetime of 10 hours for a relatively large battery due to the high power consumption of the sensor in its active mode (49 mA).

Finally, work in [15] is the only that discusses the self-discharge of the battery. This is an important problem for such long-life systems. In fact, in this work lifetime is reduced from 5.4 years to 3.7 years when accounting for the self-discharge of the 33 mAh battery.

B. ENERGY HARVESTING ARCHITECTURES

To eliminate the need for batteries, some solutions aim to achieve fully batteryless systems by incorporating energy harvesting modules, as shown in Table 1: photovoltaic [4], [9], [13], [16], [17], [20], thermoelectric [21], [23], and radio frequency [4], [6], [7], [12], [14], [22]. A significant number of solutions employ majority of photovoltaic modules. This is logical, since photovoltaic cells provide high output voltage and the highest power density in outdoor applications [4].

The architectures of the energy harvesters employed in the SoA of UHF semi-passive sensor tags are shown in Fig. 4, Fig. 5, and Fig. 6. Specifically, Fig. 4 shows the main blocks required for an RF harvester. This type of harvester collects energy from the incoming RF signals at the frequency band for which the harvester has been designed. In the case of the RF harvesters within the SoA, they are designed for the same UHF band and therefore collect energy from the same incoming RF signal that energizes the tag IC. In fact, the RF-DC rectifier of the RF harvester uses the same concept of the VM block within the tag IC. The only difference is that this RF-DC rectifier is optimized in terms of its conversion efficiency for the expected load. Moreover, high-performance Schottky diodes and high-Q passive components are used for the RF-DC rectifier and the matching network, respectively, as in [6], and [7]. Furthermore, antennas with better performance are usually employed as in [12] with an 8.7 dBi patch antenna. Still, like to the tag IC, the RF harvester also exhibits a limited sensitivity, i.e., the minimum input power that the harvester needs to operate, and must be considered for the communication range calculations of the entire semi-passive sensor tag. In general, the RF harvester's sensitivity outperforms the sensitivity of the tag IC in fully passive mode. However, this may not be true when compared to the tag IC sensitivity in BAP mode. As a consequence, the sensitivity of the RF harvester tends to be the bottleneck for the sensitivity of the sensor tag and therefore for the forward communication range. Several examples are detailed in subsection IV-C. As also shown in Fig. 4, a step-up DC-DC converter is normally required, since the output voltage provided by the RF harvester usually falls below the input voltage range accepted by the tag IC. The drawback is that the step-up converter, although high, also has a limited efficiency, which reduces the amount of harvested power, as shown in subsection IV-C. Finally, if the RF energy harvester is intended to power the

MCU and sensors, corresponding voltage regulators become necessary. In general, direct connection to the tag IC is feasible, provided that the recommended input voltage ranges are adhered to.



FIGURE 4. Architecture of an RF energy harvester.

Furthermore, Fig. 5 shows the blocks employed for the TEG or the indoor PV harvesters. In these cases, an RF-DC rectifier is not required as the output of these harvesters is already a DC signal. Again, in these cases the output voltage is typically below the input voltage range required by the tag IC for the low temperature gradients in TEGs or the indoor light intensities in PV cells. Therefore, a step-up DC-DC converter is also necessary is essential despite the associated energy losses due to the converter's limited efficiency. Subsection IV-C provides additional performance insights from the SoA. Again, if the PV or TEG harvester aims to power the MCU and sensors, corresponding voltage regulators are required.



FIGURE 5. Architecture of a TEG or an indoor PV energy harvester.

Finally, Fig. 6 shows the case of a PV harvester for outdoor conditions. The higher light intensity results in a greater output voltage, eliminating the need of a step-up DC-DC converter. Therefore, the harvester can be connected directly to the tag IC as long as the IC's input voltage range is respected by using a voltage limiter. Additionally, a diode connected series is also required to prevent any possible discharging of the tag, as exemplified in [16], [17] or [20].

C. ENERGY HARVESTING PERFORMANCE

Table 4 presents the performance of the harvesters from the works that provide sufficient details. In Table 4 "Average Power Consumption" is the power consumed by the sensor tag. "Area" refers to the area occupied by the harvesting block. "V_{OUT}" is the voltage at the output of the harvesting block, either at the output of the RF-DC rectifier or the input of the step-up converter. "P_{OUT}" is the output power at the output of the step-up converter, so its efficiency is considered.



FIGURE 6. Architecture of an outdoor PV energy harvester.

Finally, "Power Density" is the ratio of output power to the occupied area. Thus, Table 4 allows a comparison among the different energy harvester solutions. However, this comparison must be made with caution considering the differences of energy sources, conditions, energy availability and target applications.

As mentioned before, photovoltaic cells provide the greatest output power along with the highest power density in outdoor conditions as in [4] and [17]. Moreover, as mentioned before, given the high output voltage no boost converter is required. Different results are obtained when the photovoltaic modules are used in indoor applications [40], [41]. As shown in Table 4, in [20], for a typical indoor light intensity of 800 lux, a module with a 5.5 cm^2 area, provides a final output power of 35 μ W and 470 μ W with output voltage of the PC cell of 0.4 V and 0.73 V depending on the type of light source. Similarly, in [17], for a light intensity of 100 lux from fluorescent light source and output power of 20 μW is harvested. These results can be compared with energy harvesters based on TEG. For example, in [21] the TEG module with an area of 54.4 cm^2 , provides an output power of 90 μ W for 2.5°C of temperature gradient. Therefore, TEG output power levels are noticeably lower than those offered by PV technology in outdoor conditions but are comparable to power densities of PV harvesters in certain indoor applications. Still, output voltage levels are much lower if compared to indoor PV harvesters, below 100 mV as shown in [21] or [23]. Although this is due to the low temperature gradients used in these works, such a low output voltage requires special low start-up voltage step-up converters that impact the efficiency of the energy source [42]. Finally, it is also worth noting that the output power and consequently the power density of TEGs has a quadratic increment with temperature gradient [43], [44]. Therefore, higher temperature gradients would bring TEG harvesters closer to indoor PV counterparts both in terms of output power and output voltage levels.

Lastly, Table 4 presents the findings for RF harvesting from [4], [6], [7], [12], and [22]. The last column shows the harvesting frequency band of these works. As mentioned, the RF harvesters within the SoA are designed for the same UHF band that powers the tag IC. This column also shows the

Ref.	Energy Source	Avg. Pow. Cons. (µW)	Area (cm ²)	Vout (V)	Pout (µW)	Pow. Dens. (µW/cm ²)	Test conditions
[4]	PV	-	54	3.60	180000	3333	Full Sun
[4]	\mathbf{PV}	-	54	3.60	41000	759	¹ / ₄ Sun
[17]	PV	20	2	-	20*	10	Light intensity: 100 lux Light source: Fluorescent
[17]	PV	20	0.02	-	30	1500	Full Sun
[17]	PV	20	2.8	3.20	10000	3570	Full Sun
[20]	PV	31	5.5	0.40	35	6.3	Source: LED/Fluorescent Light intensity: 800 lux Boost conv. Efficiency: 70%
[20]	PV	31	5.5	0.73	470	85	Source: Halogen Light intensity: 800 lux Boost conv. Efficiency: 70%
[21]	TEG	34.4	54.4	0.037	90	1.7	Temperature gradient: 2.5°C Boost Conv. Efficiency: 22%
[23]	TEG	4.4	5.6	0.060	11	2.0	-
[4]	RF	-	-	0.35	-	-	Sensitivity: -13.8 dBm Frequency: 925 MHz
[6]	RF	-	-	0.35	-	-	Sensitivity: -14 dBm Frequency: 866 MHz
[7]	RF	414	-	0.35	25	-	Sensitivity: -9 dBm Frequency: 866.5 MHz Overall Efficiency: 20%
[12]	RF	-	-	2.1	79*	-	Sensitivity: -8 dBm Frequency: 866 MHz RF-DC rect. Efficiency: 50%
[22]	RF	-	-	0.40	11	-	Sensitivity: -15 dBm Frequency: 915 MHz RF-DC rect. Efficiency: 50% Boost conv. Efficiency: 70%

TABLE 4. Energy harvesters in semi-passive sensor tags.

* Before boost converter

sensitivity of the RF harvester. As also mentioned, although works in the SoA show good sensitivity results for the RF harvester, these results are lower than the sensitivities of the majority of tag ICs in the BAP mode. Therefore, the sensitivity of RF harvesters can become a limiting factor for the entire tag's sensitivity and therefore for the forward communication range. This is the case of the system in [4] that uses the Monza X 8K tag IC with a sensitivity of -24 dBm in BAP mode, while the RF harvester sensitivity is -13.8 dBm. Similarly, [6] and [7] use the Monza X 2K IC with -24 dBm BAP sensitivity, but the RF harvesters have sensitivities of -9 dBm and -14 dBm, respectively. The work in [22] uses the Rocky100 IC. This tag IC also has a BAP sensitivity of -24 dBm, while the sensitivity of its RF harvester is only -15 dBm. Finally, work in [12] does not include a boost converter but requires a rectified output voltage of 2.1 V. This is achieved with an input voltage of -8 dBm for a 56 k Ω load [45]. This work uses the SL900A tag IC with a sensitivity of -15 dBm in BAP mode. Therefore, the RF harvester also limits the communication range in this case. A more detailed study of the SoA performance in terms of the communication range is discussed in Subsection V-C.

Concerning the output power of RF harvesters, the data presented in [7] is calculated based on the RF power at the input and the overall system efficiency, which includes both the RF-DC rectifier and the DC-DC boost converter in this case. For an input power of -9 dBm and 20% of efficiency, the calculated harvested power is 25 μ W. In [12] an input power of -8 dBm and an efficiency of 50% for the RF-DC rectifier corresponds to 79 μ W of output power. In [22] only the efficiency of the RF-DC rectifier is provided but not the overall efficiency. However, since the system employs a commercial boost converter, the harvested power can be estimated based on the boost efficiency at the specific voltage provided by the rectifier output [46]. It results in 11 μ W for -15 dBm input power.

Concerning power density of these solutions, unfortunately it could not be calculated as antenna dimensions are not



FIGURE 7. Semi-passive sensor tag architecture with externally supplied tag IC.

provided, although these results of these works are within the SoA of RF harvesters [47], [48].

Finally, to ensure that the generated energy is sufficient to meet the energy demands of the sensor tag and to develop a battery-free sensor tag, it is essential to conduct an energy budget analysis. This analysis involves considering the supply voltage required by the tag's components such as the tag IC, MCU, and sensors, as well as their active and sleep current consumption. Furthermore, for a comprehensive analysis, the availability of the energy source must also be considered. By utilizing these values, it becomes possible to calculate the duty cycle of the sensor tag, which determines the sensor measurement rate and consequently the average system power consumption. Several works in the SoA of semi-passive sensor tags with energy harvesters discuss an energy budget analysis as in [7], [17], [21], and [23].

V. SEMI-PASIVE SENSOR TAGS: ARCHITECTURES AND PERFORMANCE

This section discusses the architectures and performance of the whole sensor tags in the SoA. The discussion starts with the architectures employed in the SoA. In addition, as the most considered parameter of semi-passive tags is the read range, this section discusses the theoretical forward link read range of the tag ICs. This helps analyze the performance of sensor tags within the SoA in terms of the read range.

A. ARCHITECTURES OF SENSOR TAGS

The state-of-the-art for semi-passive sensor tags utilizes two main architectures, as illustrated in Fig. 7 and Fig. 8. Both architectures incorporate a low power MCU that communicates with the tag IC via a first SPI or I2C port. The MCU is also responsible for communicating with a digital sensor through a second SPI or I2C port or, in some cases, it digitizes the output of an analog sensor by means of an ADC. As mentioned before, these architectures offer the advantage of data logging and high flexibility since any digital or analog sensor can be accommodated in practice. However, there are



FIGURE 8. Semi-passive sensor tag architecture with passive tag IC.

exceptions to these architectures. For example [11], [14], [20], [27], and [29] directly connect their sensors to the tag IC without using a low-power MCU. In general, these sensor tags employ the internal ADC to digitize an analog sensor. Other works, such as [12], [16], [17], and [26] use only the available internal temperature sensor. In [9], an analog sensor is part of the matching network between the antenna and the tag IC, so the sensing mechanism is based on the minimum transmit power required to activate tag IC. In [24] although the sensor is externally supplied, the sensing mechanism is integrated in the backward communication protocol. It does not make use of an MCU, but it does not fully follow the EPC Global G2/ISO-18000C communication standard. Finally, there is no solution utilizing digital sensors connected directly to the tag IC although it is possible for the tag ICs with an SPI Master module.

The architecture depicted in Fig. 7 is the most commonly used among the SoA solutions. It is employed in [3], [4], [8], [10], [18], [19], [21], and [22]. As shown in Fig. 7, the battery or energy harvesting module supplies power to all the devices: the tag IC, the MCU and the sensor. Since they use class 3 tag ICs, once the tag IC is externally powered, it can utilize the BAP mode for extended communication range.

However, this is not the only architecture adopted in the SoA. Fig. 8 presents a different approach. It also employs an intermediate MCU for communication with digital sensors, but the main difference is that the battery/energy harvester supplies power to the MCU and the sensors, but not the tag IC. As previously mentioned, this is an acceptable solution since the sensitivity of the tag IC remains unaffected despite the power consumption of the MCU and sensors, as these are powered by the battery or energy harvester module.

This architecture is used in [3], [6], [7], [15], [23], [25], and [28], and it has been chosen in these SoA solutions for two primary reasons. The first reason is related to the characteristics of the battery or the energy harvester module. For instance, in [15] a paper battery with limited capacity is employed, so the sensor tag is kept in passive mode for autonomy extension. In [6] and [7], RF harvesting is employed with

Ref	Manufacturer	Model	Passive Sensitivity (dBm)	BAP Sensitivity (dBm)	Theoretical Passive FW range (m)	Theoretical BAP FW range (m)
[31]	Impinj	Monza X 8K	-19.1	-26.1	12.6	28.4
[32]	Impinj	Monza X 2K	-17	-24	10	22.3
[33]	AMS	SL900A	-7	-15	3.1	7.9
[34]	EM Micro.	EM4325	-8.3	-17/-31	3.7	10/50
[35]	Farsens	Rocky100	-14	-24/-35	7	22.3/79
[36]	Farsens	ANDY100	-4	-	2.2	-
[37]	NXP	G2iL	-17.6	-27	10.6	31.4
[38]	Fujitsu	MB97R804B	-6	-	2.8	-

TABLE 5. Tag IC read range performance.

a sensitivity is similar to that of the tag IC in passive mode. Consequently, utilizing the tag IC in BAP mode becomes impractical, as the energy harvesting module becomes the limiting factor for communication range. In fact, in [7] a test using BAP mode is also carried out, showing greater communication range but, in this case, a stable DC source seems to be employed.

The second reason is simply that the tag IC lacks BAP mode, as it is the case in [23], [25], and [28]. In such cases, enhanced sensitivity is not possible.

In conclusion, to achieve an extended communication range, the BAP mode of the tag IC is desirable. However, it is important to also consider the type of energy harvester being used. For example, in the case of RF harvesting, the limited sensitivity of the harvester itself can make the BAP mode ineffective. Nevertheless, even if the harvested energy is insufficient to enable the IC's BAP mode, it does not necessarily make the harvester useless. By means of the second architecture, the role of the battery/energy harvester is to maintain the tag IC's sensitivity unaffected by load conditions. In this way, the reading range is improved.

B. SENSOR TAG READ RANGE

Table 5 shows the sensitivity and the communication range of the different tags IC adopted in the SoA based on the published sensitivities (usually measured with an ID query command) and calculated with (1).

$$P_{IC} = \left(\frac{\lambda}{4\pi d}\right)^2 P_{Tx} G_{Tx} G_{Tag} \eta \tau \tag{1}$$

Equation (1) computes the power reaching the input ports of the tag IC after the sensor tag antenna, where P_{IC} is the input power, P_{TX} the reader transmitted power and G_{TX} and G_{Tag} are the reader and antenna gains, respectively. In addition, η refers to the polarization losses and τ the antenna-to-tag mismatch. Finally, $(\lambda/4\pi d)^2$ refers to the free space losses. For the read range calculations, a reader with output power of 2W ERP (complying with the European UHF band power regulations), a 6 dBic gain circular polarized reader antenna and 2 dBi gain dipole sensor tag antenna are assumed. Perfect matching between the antenna and the sensor tag is considered for simplicity.

As mentioned earlier, class 3 tag ICs exhibit different sensitivity performance when configured in passive or BAP modes. This is shown in Table 5 when comparing the columns for the passive range and the BAP range. As observed, a 7 dB to 10 dB sensitivity improvement is achieved when the tag IC is configured in BAP mode. As a consequence, the read range is doubled or tripled.

Among the class 3 tag ICs, the tag IC in [35] achieves a remarkable performance. It has two distinct BAP sensitivities: BAP and enhanced BAP or E-BAP. In BAP mode it shows a sensitivity of -24 dBm with a theoretical reading range of 22.3 m. This sensitivity is similar to other tag ICs, and it is the choice of the solutions in the SoA. However, in E-BAP mode, the tag IC achieves a sensitivity of -35 dBm corresponding to a theoretical reading range of 79 m. Another example is [34], whose sensitivity in BAP mode ranges from -17 dBm to -31 dBm (it also allows intermediate values of -22 dBm and -28 dBm), so for a 2 W ERP reader and 2 dBi sensor tag antenna gain, its theoretical read range extends from 10 m to 50 m.

It should be noted that the read range shown in Table 5 represents only the communication range of the forward link, (reader-to-tag). The overall system communication range is not fully characterized until the backward link (tag-to-reader) is also analyzed, although in the SoA only a few papers discuss the backward link [16], [21]. In passive mode, the tag IC sensitivity and hence the forward link is clearly the limiting factor for the overall read range. Conversely, in semi-passive mode, the sensitivity of the reader in the backward link can become the bottleneck.

To illustrate this, Fig. 9 shows the signal power reaching both the input of the tag IC and the backscattered power at the reader side. The calculations are based on (1) and (2) [49].

$$P_{Rec} = \left(\frac{\lambda}{4\pi d}\right)^4 P_{Tx} G_{Tx}^2 G_{Tag}^2 \eta^2 k \tag{2}$$

Equation (2) provides the power received by the reader considering both the reader-to-tag and tag-to-reader communication links. In Equation 2, factor k represents the modulation factor, i.e., the losses associated with the backscattering mechanism of the tag.

Fig. 9 illustrates the P_{IC} and P_{REC} for the European power regulations at the UHF band: 2 W ERP at 868 MHz. Again, it considers a 6 dBic gain circular polarized reader antenna and 2 dBi gain dipole sensor tag antenna and a modulation loss factor of -6 dB [50]. Finally, to determine the forward

TABLE 6. Sensor tag configuracions and communication range limits.

Sensor Tag Configuration	Description	Sensitivity	Range limit
Passive/MCU/Sensors (Fully passive sensor tag)	Passive sensor tag in load conditions RF signal energizes tag IC and external devices	Less than the passive query command sensitivity	Forward link Energy limited
Passive/Battery-Harvester/MCU/Sensors (Semi-passive sensor tag/tag IC in passive mode)	Passive sensor tag in no-load conditions RF signal energizes only tag IC Battery-Harvester supplies external devices	Equals the passive query command sensitivity	Forward link Energy limited
BAP/Battery-Harvester/MCU/Sensors (Semi-passive sensor tag/tag IC in BAP mode)	Battery-Harvester supplies both tag IC and external devices	Enhanced sensitivity	Forward link or backward link- Communication limited



FIGURE 9. Theoretical forward and backward link communication range for tag IC and reader sensitivities of -24dBm and -80dBm, respectively.

and backward link communication ranges, sensitivities of -24 dBm and -80 dBm are assigned to the tag IC and the reader, respectively. This last value can be found in SoA for reader sensitivities [16]. As observed in Fig. 9, the forward link offers a potential range of 22 m for 2 W ERP and a tag IC sensitivity of -24 dBm. However, in this case, the overall read range would be limited to 18m when considering the backward link as well. Of course, it is possible to extend the overall communication range by increasing the gain of the reader antenna. Fig. 10 shows how the backward link communication range expands with higher reader gains. However, a drawback in this case is that the antenna becomes more directional, which may be a problem for applications that requires omnidirectional reading. Alternatively, recent works implement reflected signal amplification to extend the backward link communication range but at the cost of higher power consumption of the sensor tags [51].

As a summary of the previous discussions, Table 6 collects the communication range limits across different scenarios. In a fully passive configuration, which corresponds to Fig. 1a,



FIGURE 10. Theoretical communication range vs gain of the antenna reader and reader sensitivity of -80dBm.

the communication range is energy constrained due to the demanding load conditions of the tag IC, since the energy harvested from the reader energizes not only the tag IC but also the external MCU and sensors. In such cases, the sensitivity of the tag IC is degraded compared to the level shown in datasheets, which is measured using query commands for ID reading. Consequently, the forward link becomes the bottleneck in such scenarios.

Alternatively, there is a first semi-passive configuration when the battery or the energy harvester module supplies power to the MCU and the sensors but not the tag IC, as in Fig. 8. Therefore, the incoming energy from the reader is used only to activate tag IC. Therefore, the tag IC works under noload conditions, and, in this case, the expected sensitivity is the one provided by the manufacturer in passive mode. In this case, the range is constrained by the forward link, provided that the reader has an acceptable sensitivity.

The second semi-passive configuration case occurs when the battery or the energy harvester supplies power not only to the MCU and the sensors but also to the tag IC configured in BAP mode, as in Fig. 7. Here, the system is not energy limited but communication limited since it is the backward

TABLE 7.	Sensor tag	configura	cions and	communication	range limits.
----------	------------	-----------	-----------	---------------	---------------

Ref	IC Supplier	IC Model	Energy Source	Fully Passive Range (m)	Passive/No load Range (m)	BAP Range (m)	Tag antenna gain (dBi)	Theoretical Passive range (m)	Theoretical BAP range (m)
[3]	Impini	Monza X 8K	Battery		11	18	g	12.6	28.4
[4]	Impini	Monza X 8K	RF/PV	-	-	7 48/27	5 98	12.6	28.4
[5]	Impini	Monza X 8K	Battery	-	-	1.5		12.6	28.4
[6]	Impini	Monza X 2K	RF	1.5	5	-	-	10	22.3
[7]	Impini	Monza X 2K	RF(Battery)	-	10	-	1.8	10	22.3
[8]	Impini	Monza X 2K	Battery	-	_	22	2.4	10	22.3
[9]	Impini	Monza X	PV/Battery	2.4	-	4.6/5.4	4.8	-	-
[10]	AMS	SL900A	Battery	1.8	-	2.8	1	3.1	7.9
<u>î</u> 11	AMS	SL900A	Battery	1	-	2.8	-10	3.1	7.9
[12]	AMS	SL900A	RF	-	-	3	8	3.1	7.9
[13]	AMS	SL900A	PV	3.6	-	7.3	-	3.1	7.9
[14]	AMS	SL900A	RF	-	-	1.13	-	3.1	7.9
[15]	EM Micro.	EM4325	Battery	-	3.4	-	-	3.7	10/50
[16]	EM Micro.	EM4325	PV	1	-	7	-	3.7	10/50
[17]	EM Micro.	EM4325	PV	1	4	-	-	3.7	10/50
[18]	EM Micro.	EM4325	Battery	-	-	9	1.52	3.7	10/50
[19]	EM Micro.	EM4325	Battery	-	-	4	-	3.7	10/50
[20]	Farsens	Rocky100 IC	PV	3.7	-	21	2.1	7	22.3
[21]	Farsens	Rocky100 IC	TEG	4.4	-	22	2.1	7	22.3
[22]	Farsens	Rocky100 IC	RF	-	-	8.5	1.16	7	22.3
[23]	Farsens	ANDY100	TEG	1.75	3	-	2	2.2	-
[24]	NXP	G2IL	Battery	-	-	20	2	10.6	31.4
[25]	Fujitsu	MB97R804B	Battery		0.731	-	-	2.8	-
[26]	-	Custom	Battery	-	-	-	-	-	21.5
[27]	-	Custom	Battery	-	-	1	0	-	-
[28]	-	Custom	Battery	6*	2.7*	-	-	5.1	2.3
[29]	-	Custom	Battery	-	-	4.3	2	-	3

link the one that becomes the bottleneck due to limitations of the reader sensitivity.

C. SoA READ RANGE

Table 7 presents the actual communication range of the SoA solutions. Based on the previous discussion Table 7 distinguishes the three different previously mentioned sensor tag configurations: Fully Passive Range when the tag IC is in passive mode and there is no extra energy source attached to the sensor tag. Passive/No-load Range when the sensor tag uses an energy harvester, but the tag IC is in passive mode. BAP Range when the sensor tag uses an energy harvester, and the tag IC is in BAP mode.

As shown in Table 7, the read ranges exhibit a wide variation. Moreover, some solutions show a significant discrepancy when compared to the theoretical maximum read range in the last two columns. This theoretical read range is again based on the tag IC's sensitivities provided by the tag IC manufacturers and considering 2 W ERP transmission power, a 6 dBic gain for reader antenna and a 2 dBi gain for the sensor tag antenna. The main reasons for the discrepancies are discussed through the next subsections:

1) USE OF THE TAG IC IN PASSIVE MODE

As mentioned before, there are several cases in which the tag IC is not configured in BAP mode. Therefore, the sensitivity of the tag IC is that of the passive mode. However, as it was also previously discussed, the sensitivity is not deteriorated due to the assistance of the battery/energy harvester. For

VOLUME 11, 2023

example, in [23], the tag IC does not have BAP mode, but the authors demonstrate a range improvement from 1.75m to 3m when comparing the fully passive configuration to the semi-passive configuration. As shown, the reported read range is similar to the theoretical range for the tag IC in passive mode, which illustrates the read range improvement in no-load conditions.

In [25] the case is similar, as the tag IC only works in passive mode so that the battery is only used to power the MCU and the sensors. In addition, a dipole antenna placed on a metal surface is used, degrading the antenna performance and consequently the read range. However, it is important to note that in this case, there is no intention to optimize the range. In fact, antenna performance degradation seems to be a goal to reduce the cross talk between nearby tags.

In [17], although the tag IC can be configured in BAP mode, it seems to be in passive mode. However, the read range is improved when fully passive and semi-passive configurations are compared. Again, the reported read range is similar to the theoretical range for the tag IC in passive mode and illustrates the read range improvement in no-load conditions.

Other examples are works [6] and [7] that use UHF RF energy harvesting to power a sensor tag. For the reasons mentioned earlier, the energy harvester only supplies power to the MCU and the sensor, so the tag IC operates in passive mode. This explains the reduction in the reading range. In fact, [7] claims a read range of 22 m when the tag IC is in BAP mode using a battery as a stable power source. However, the read range in [7] is the same as the range theoretically calculated. In [6] the read range is reduced to a half, from 10 m to 5 m. A lower gain of the sensor tag antenna, and the fact that experiments were conducted in an office room, could explain it.

A last example is found in [15], where authors keep the sensor tag in passive mode to increase the autonomy of the batteries. In this work low-capacity printed batteries are used.

2) SENSITIVITY OF THE RF ENERGY HARVESTER

The characteristics of the energy harvester can contribute to a reduced sensor range. For example, the system in [4] utilizes both RF and photovoltaic harvesting and BAP mode for the tag IC. With the photovoltaic harvester the read range reaches 27 m, which is closed to the theoretical one in BAP mode, but it drops to 7.48m when only the RF harvester is employed, as the RF harvester's sensitivity of -13.8 dBm limits the overall communication range. It should be noted that authors report a 6 dBi gain for the sensor tag antenna, which should increase the read range beyond the theoretical calculated one. However, the authors also report a loss in antenna performance due to the loading effect of the solar panel, which is placed on top of the antenna.

Work in [12] utilizes the tag IC in BAP mode and adds UHF RF harvesting. However, the authors state a reading range of 3 m, which is lower than the potential range of 13 m calculated by authors (8 dBi linearly polarized sensor tag antenna is used) or the theoretical read range in BAP mode of 7.9 m in Table 7. Again, this is due to the sensitivity constraints of the harvester. As it was shown in Table 4 the RF harvesting sensitivity is -8 dBm input voltage for a 56 k Ω load [45].

Similarly, [22] utilizes UHF RF Harvesting and although the tag IC is in BAP mode with enhanced sensitivity of -24 dBm, the overall sensitivity of the sensor tag is also limited by the RF harvester's sensitivity of -12.5 dBm, resulting in a reduced read range.

3) LIMITATIONS DUE TO THE APPLICATION

In certain cases, the application itself imposes constraints on the read range due to factors such as multipath effects, the influence of surroundings on antenna performance, or limitations related to antenna size.

For example, in [11] an open-loop antenna designed to be placed on the human body exhibits a gain of -10 dBi, which consequently reduces the potential read range. In [24] the reduced read range is a consequence on the testing conditions, with the authors reporting a 20 m read range in indoor environment and therefore affected by multipath effects.

In [14] the designed sensor tag shows a read range of only 0.98 m in passive mode and 1.13 m with a RF energy harvester. The authors attribute this limitation to the insufficient power reaching the reader and can be caused by the negative effect of the human body on the tag's antenna performance, since the antenna seems to be a dipole and the sensor tag is being placed near the chest.

A similar underlying reason accounts for reduced read performance in [16], [18], and [19]. In [16] authors present two different solutions. One is an object tracker, where no sensor is involved. However, the second solution is a system to monitor buildings. In the latter case, a dipole antenna is placed on windows, which affects the performance of the antenna. Moreover, the authors state that the sensor tag antenna is not optimized for the effect of surrounding dielectric materials. Still, the reported read range of 7 m is not far from the minimum theorical read range of 10 m, as shown in Table 7. Similarly, works in [18] and [19] use a PIFA and a ceramic antenna, respectively, attached to the body. These types of antennas are negatively affected by the human body. In fact, both report a read range reduction when attached to the body compared to the sensor tag in free space. In [18] the authors report a read range degradation of 4 m, from 13 m to 9 m, although this is close to the theoretical minimum range of 10 m. In [19] the read range exhibits a stronger degradation from 17 m to 4 m.

4) LIMITATIONS OF THE READER

In some cases, the performance limitations of the reader contribute to a reduction in the read range. The work in [3] reports 1.8m reading range using Monza X 8K with a theoretical reading range of 28.4 m. There is limited information available on the possible causes of this reduced range. One possible reason is the fact that the authors utilize of a reader with a maximum transmit power of 23 dBm, a 5.5 dBic antenna and a -65 dBm receive sensitivity. Under these conditions the transmitted power falls well below the maximum 2 W ERP allowed in Europe (a transmit power of 29.65 dBm for a 5.5 dBic antenna gain is possible). Moreover, the sensitivity of the reader also affects the detection of the sensor tag response in backward communication, as it is 15 dB lower than the sensitivities observed in SoA readers [16]. In fact, a theoretical calculation based on the transmission power and the sensitivity of its reader results in 5 m read range, limited by the sensitivity of the reader.

Similarly, in [10] the reduced range can be partially attributed to a combination of low gain of the sensor tag antenna and low transmitted power of 26 dBm with a 7 dBi gain of the reader antenna (a transmit power 25.15 dBm for a 7 dBic antenna is possible). This combination results in a theoretical read range of 5.5 m.

5) LIMITATIONS OF THE SENSING MECHANISM

A final example is found in [9], where the sensing mechanism intentionally introduces a mismatch between the tag IC and the antenna. Although this approach is a simple solution for resistive sensors, it reduces the reading range. Additionally, in this particular case, an extra reference tag IC is required.

Lastly, in [5], there is no study of maximum reading range. The reported 1.5 m is only the test setup conditions.

As a conclusion to this section, to achieve the longest communication range for a semi-passive sensor tag, the tag IC should be configured in BAP mode. Still, keeping the tag IC in passive mode can be an acceptable solution if an RF harvester with limited sensitivity or a battery with low capacity is employed. Logically, configuring the tag IC in passive mode will result in a reduced communication range compared to the BAP mode, but will improve the communication range of a fully-passive solution. Additionally, the communication range is also affected by the influence of environmental factors such as multipath effects and antenna performance due to surrounding materials. Finally, given the high sensitivity of the tag ICs the overall communication range depends on both the forward and the backward links. Therefore, the performance of the reader must be also assessed, since the backward link can become the bottleneck of the overall communication range.

VI. CONCLUSION

This paper provides a comprehensive overview and analysis of the state-of-the-art in EPC Global G2/ISO-18000C compliant semi-passive UHF RFID sensor tags. The research highlights the significant interest in semi-passive or BAP sensor tags within the IoT industry, owing to their improved performance compared to fully passive tags. By incorporating a battery or energy harvesting module, semi-passive sensor tags extend the communication range and support power-demanding sensors and low-power MCUs without compromising communication capabilities. The paper thoroughly examines recent advancements in semi-passive UHF RFID sensor tags, evaluating crucial aspects such as application suitability, sensing capabilities, tag IC performance and features, and system communication range. The key conclusions drawn from this review are first that the tag IC include different features so that a careful analysis is crucial for optimal tag IC selection. For example, an application that requires an orientation-free solution may be interested in [31] or [32], whereas [33] is a proper option if an analog sensor is used and no MCU is to be avoided. Secondly, the overall communication range is not only limited by the tag IC's sensitivity but also by the performance of the energy harvester. As an example, the best sensitivity among the tag ICs is [35] with a published value of -35 dBm in E-BAP mode. This high sensitivity contrasts with the best sensitivity among the RF harvesters, which is found to be -15 dBm. In addition, the remarkable sensitivities of class 3 tag ICs in BAP mode also require analyzing the backward communication link so, for example, readers with a sensitivity lower than -80 dBm will limit the overall communication range. Finally, the external energy source employed for the sensor tag needs to be evaluated. Apart from a battery, which is a stable energy source, different energy harvester modules have been found within the SoA, such as PV, RF or TEG. The amount energy provided by these sources is very different. For example, a TEG in [21] is only able to provide 90 uW and 37 mV with a modest conversion efficiency, while a PV for an outdoor application provides 180 mW and 3.6 V as in [4] for a similar area, but the availability of these sources is different too. These aspects must be considered to assess the sensing rate along with the choice of the MCU or the sensors of the sensor tag.

REFERENCES

- H. A. Abdulghani, N. A. Nijdam, and D. Konstantas, "Analysis on security and privacy guidelines: RFID-based IoT applications," *IEEE Access*, vol. 10, pp. 131528–131554, 2022.
- [2] FENIX-RM, Farsens SL, Shenzhen, China, Accessed: May 26, 2023. [Online]. Available: https://www.richrfid.com/epc-c1g2-batterylessambient-temperature-sensor.html
- [3] D. Jayawardana, S. Kharkovsky, R. Liyanapathirana, and X. Zhu, "Measurement system with accelerometer integrated RFID tag for infrastructure health monitoring," *IEEE Trans. Instrum. Meas.*, vol. 65, no. 5, pp. 1163–1171, May 2016.
- [4] A. E. Abdulhadi and R. Abhari, "Multiport UHF RFID-tag antenna for enhanced energy harvesting of self-powered wireless sensors," *IEEE Trans. Ind. Informat.*, vol. 12, no. 2, pp. 801–808, Apr. 2016.
- [5] D. Jayawardana, R. Liyanapathirana, and X. Zhu, "RFID-based wireless multi-sensory system for simultaneous dynamic acceleration and strain measurements of civil infrastructure," *IEEE Sensors J.*, vol. 19, no. 24, pp. 12389–12397, Dec. 2019.
- [6] D. De Donno, L. Catarinucci, and L. Tarricone, "Enabling self-powered autonomous wireless sensors with new-generation I2C-RFID chips," in *IEEE MTT-S Int. Microw. Symp. Dig.*, Seattle, WA, USA, Dec. 2013, pp. 1–4.
- [7] D. De Donno, L. Catarinucci, and L. Tarricone, "RAMSES: RFID augmented module for smart environmental sensing," *IEEE Trans. Instrum. Meas.*, vol. 63, no. 7, pp. 1701–1708, Jul. 2014.
- [8] D. De Donno, L. Catarinucci, and L. Tarricone, "A battery-assisted sensorenhanced RFID tag enabling heterogeneous wireless sensor networks," *IEEE Sensors J.*, vol. 14, no. 4, pp. 1048–1055, Apr. 2014.
- [9] A. E. Abdulhadi and T. A. Denidni, "Self-powered multi-port UHF RFID tag-based-sensor," *IEEE J. Radio Freq. Identificat.*, vol. 1, no. 2, pp. 115–123, Jun. 2017.
- [10] J. F. Salmerón, F. Molina-Lopez, A. Rivadeneyra, A. V. Quintero, L. F. Capitán-Vallvey, N. F. de Rooij, J. B. Ozáez, D. Briand, and A. J. Palma, "Design and development of sensing RFID tags on flexible foil compatible with EPC gen 2," *IEEE Sensors J.*, vol. 14, no. 12, pp. 4361–4371, Dec. 2014.
- [11] C. Miozzi, S. Nappi, S. Amendola, C. Occhiuzzi, and G. Marrocco, "A general-purpose configurable RFID epidermal board with a twoway discrete impedance tuning," *IEEE Antennas Wireless Propag. Lett.*, vol. 18, no. 4, pp. 684–687, Apr. 2019.
- [12] S. Ma, N. Pournoori, L. Sydänheimo, L. Ukkonen, T. Björninen, and A. Georgiadis, "A batteryless semi-passive RFID sensor platform," in *Proc. IEEE Int. Conf. RFID Technol. Appl. (RFID-TA)*, Pisa, Italy, Sep. 2019, pp. 171–173.
- [13] A. Abdelnour, A. Hallet, S. B. Dkhil, P. Pierron, D. Kaddour, and S. Tedjini, "Energy harvesting based on printed organic photovoltaic cells for RFID applications," in *Proc. IEEE Int. Conf. RFID Technol. Appl. (RFID-TA)*, Pisa, Italy, Sep. 2019, pp. 110–112.
- [14] A. Rigi, A. J. Mugisha, A. Arefian, S. R. Khan, and S. Mitra, "Wireless battery-free body temperature sensing device for key workers," *IEEE Sensors Lett.*, vol. 6, no. 2, pp. 1–4, Feb. 2022.
- [15] T. Unander, J. Siden, and H.-E. Nilsson, "Designing of RFID-based sensor solution for packaging surveillance applications," *IEEE Sensors J.*, vol. 11, no. 11, pp. 3009–3018, Nov. 2011.
- [16] S. N. R. Kantareddy, I. Mathews, R. Bhattacharyya, I. M. Peters, T. Buonassisi, and S. E. Sarma, "Long range battery-less PV-powered RFID tag sensors," *IEEE Internet Things J.*, vol. 6, no. 4, pp. 6989–6996, Aug. 2019.
- [17] S. N. R. Kantareddy, I. Mathews, S. Sun, M. Layurova, J. Thapa, J.-P. Correa-Baena, R. Bhattacharyya, T. Buonassisi, S. E. Sarma, and I. M. Peters, "Perovskite PV-powered RFID: Enabling low-cost selfpowered IoT sensors," *IEEE Sensors J.*, vol. 20, no. 1, pp. 471–478, Jan. 2020.
- [18] R. Colella, M. R. Tumolo, S. Sabina, C. G. Leo, P. Mincarone, R. Guarino, and L. Catarinucci, "Design of UHF RFID sensor-tags for the biomechanical analysis of human body movements," *IEEE Sensors J.*, vol. 21, no. 13, pp. 14090–14098, Jul. 2021.

- [19] R. Colella, L. Spedicato, C. G. Leo, S. Sabina, V. Congradac, O. Bagatin, and L. Catarinucci, "Design and technology transfer of RFID-based medical sensing devices," in *Proc. IEEE Int. Conf. RFID Technol. Appl.* (*RFID-TA*), Delhi, India, Oct. 2021, pp. 191–194.
- [20] A. Astigarraga, A. Lopez-Gasso, D. Golpe, A. Beriain, H. Solar, D. del Rio, and R. Berenguer, "A 21 m operation range RFID tag for 'pick to light' applications with a photovoltaic harvester," *Micromachines*, vol. 11, no. 11, p. 1013, Nov. 2020.
- [21] H. Solar, A. Beriain, A. Rezola, D. del Rio, and R. Berenguer, "A 22m operation range semi-passive UHF RFID sensor tag with flexible thermoelectric energy harvester," *IEEE Sensors J.*, vol. 22, no. 20, pp. 19797–19808, Oct. 2022.
- [22] A. Nadeem, D. Chatzichristodoulou, A. Quddious, N. Shoaib, L. Vassiliou, P. Vryonides, and S. Nikolaou, "UHF IoT humidity and temperature sensor for smart agriculture applications powered from an energy harvesting system," in *Proc. IEEE Int. Conf. Internet Things Intell. Syst. (IoTaIS)*, Bali, Indonesia, Nov. 2022, pp. 186–190.
- [23] I. Jauregi, H. Solar, A. Beriain, I. Zalbide, A. Jimenez, I. Galarraga, and R. Berenguer, "UHF RFID temperature sensor assisted with bodyheat dissipation energy harvesting," *IEEE Sensors J.*, vol. 17, no. 5, pp. 1471–1478, Mar. 2017.
- [24] E. DiGiampaolo, A. DiCarlofelice, and A. Gregori, "An RFID-enabled wireless strain gauge sensor for static and dynamic structural monitoring," *IEEE Sensors J.*, vol. 17, no. 2, pp. 286–294, Jan. 2017.
- [25] M. Tanaka, N. Takahashi, R. Ikeda, H. Yoda, and M. Suzuki, "Development and evaluation of a ground coil monitoring system using semi-passive sensor tags," in *Proc. IEEE Int. Conf. RFID Technol. Appl. (RFID-TA)*, Tokyo, Japan, Sep. 2015, pp. 141–146.
- [26] W. Che, D. Meng, X. Chang, W. Chen, L. Wang, Y. Yang, C. Xu, X. Tan, N. Yan, and H. Min, "A semi-passive UHF RFID tag with on-chip temperature sensor," in *Proc. IEEE Custom Integr. Circuits Conf.*, San Jose, CA, USA, Sep. 2010, pp. 1–4.
- [27] C.-P. Wang, S.-Y. Lee, and W.-C. Lai, "An RFID tag system-on-chip with wireless ECG monitoring for intelligent healthcare systems," in *Proc.* 35th Annu. Int. Conf. IEEE Eng. Med. Biol. Soc. (EMBC), Osaka, Japan, Jul. 2013, pp. 5489–5492.
- [28] S.-M. Yu, P. Feng, and N.-J. Wu, "Passive and semi-passive wireless temperature and humidity sensors based on EPC Generation-2 UHF protocol," *IEEE Sensors J.*, vol. 15, no. 4, pp. 2403–2411, Apr. 2015.
- [29] V.-H. Duong, N. X. Hieu, H.-S. Lee, and J.-W. Lee, "A battery-assisted passive EPC Gen-2 RFID sensor tag IC with efficient battery power management and RF energy harvesting," *IEEE Trans. Ind. Electron.*, vol. 63, no. 11, pp. 7112–7123, Nov. 2016.
- [30] The EPCglobal Architecture Framework, GS1 Final Version 1.5, Global Standards 1, GS1, Brussels, Belgium, Mar. 2013. [Online]. Available: https://www.gs1.org/
- [31] Impinj Monza X-8K IC Data Sheet, Impinj, Seattle, WA, USA, Accessed: May 26, 2023. [Online]. Available: https://support.impinj.com/hc/enus/articles/202756868-Monza-X-8K-Dura-Product-Brief-Datasheet?
- [32] Impinj Monza X-2K IC Data Sheet, Impinj, Seattle, WA, USA, Accessed: May 26, 2023. [Online]. Available: https://support.impinj.com/hc/enus/articles/202756848-Monza-X-2K-Dura-Datasheet?
- [33] AMS SL900A IC Data Sheet, Premstaetten, Austria, Accessed: May 26, 2023. [Online]. Available: https://ams.com/sl900a
- [34] EM4325 IC Data Sheet, EM Microelectronic, Marin, Switzerland, Accessed: May 26, 2023. [Online]. Available: https://www. emmicroelectronic.com/product/epc-and-uhf-ics/em4325
- [35] ROCKY100 IC Data Sheet, Farsens SL, Shenzhen, China, Accessed: May 26, 2023. [Online]. Available: https://www.richrfid.com/uploads/DS-ROCKY100-V04.pdf
- [36] ANDY100 IC Sheet, Farsens SL, Shenzhen, Data [Online]. China. Accessed: May 26, 2023.Available: https://community.st.com/t5/st25-nfc-rfid-tags-and-readers/send-spicommands-to-farsens-andy100-rfid-tag-using-st25ru3993/m-p/135726
- [37] UCODE G2iL IC Data Sheet, NXP Semiconductors, Eindhoven, The Netherlands. Accessed: May 26, 2023. [Online]. Available: https://www.nxp.com/products/rfid-nfc/ucode-rain-rfid-uhf/ucodeg2il-and-g2il-plus: SL3S1203_1213
- [38] MB97R804B Data Sheet, Fujitsu, Tokyo, Japan, Accessed: May 26, 2023. [Online]. Available: https://www.fujitsu.com/uk/Images/MB97R803A.pdf
- [39] UCODE G2iM+ IC Data Sheet, NXP Semiconductors, Eindhoven, The Netherlands, Accessed: May 26, 2023. [Online]. Available: https://www.nxp.com/products/rfid-nfc/ucode-rain-rfid-uhf/ucodeg2im-and-g2im-plus: SL3S1003_1013

- [40] A. Nasiri, S. A. Zabalawi, and G. Mandic, "Indoor power harvesting using photovoltaic cells for low-power applications," *IEEE Trans. Ind. Electron.*, vol. 56, no. 11, pp. 4502–4509, Nov. 2009.
- [41] I. Mathews, P. J. King, F. Stafford, and R. Frizzell, "Performance of III–V solar cells as indoor light energy harvesters," *IEEE J. Photovolt.*, vol. 6, no. 1, pp. 230–235, Jan. 2016.
- [42] LTC3108 Ultralow Voltage Step-Up Converter and Power Manager, Analog Devices Inc., Wilmington, MA, USA, Accessed: May 26, 2023. [Online]. Available: https://www.analog.com/media/en/technicaldocumentation/data-sheets/LTC3108.pdf
- [43] J. Xie, C. Lee, and H. Feng, "Design, fabrication, and characterization of CMOS MEMS-based thermoelectric power generators," *J. Microelectromech. Syst.*, vol. 19, no. 2, pp. 317–324, Apr. 2010.
- [44] M. D. Chen, J. Y. Wang, S. M. Yang, and M. H. Tsai, "Structural design for dimensional stability of thermocouples in thermoelectric energy harvester," *IEEE Sensors J.*, vol. 19, no. 1, pp. 58–64, Jan. 2019.
- [45] N. Pournoori, L. Ukkonen, L. Sydänheimo, and T. Björninen, "Charge storage level sensor RFID tag: Impedance matching and experimental characterisation," in *Proc. 13th Eur. Conf. Antennas Propag. (EuCAP)*, Krakow, Poland, Mar. 2019, pp. 1–5.
- [46] AEM30940 Data Sheet, e-PeasSA, Leuven, Belgium, Accessed: May 26, 2023. [Online]. Available: https://e-peas.com/product/aem30940/
- [47] Y. C. Lee, H. Ramiah, A. Choo, K. K. P. Churchill, N. S. Lai, C. C. Lim, Y. Chen, P.-I. Mak, and R. P. Martins, "High-performance multiband ambient RF energy harvesting front-end system for sustainable IoT applications—A review," *IEEE Access*, vol. 11, pp. 11143–11164, 2023.
- [48] A. C. C. Chun, H. Ramiah, and S. Mekhilef, "Wide power dynamic range CMOS RF-DC rectifier for RF energy harvesting system: A review," *IEEE Access*, vol. 10, pp. 23948–23963, 2022.
- [49] J. D. Griffin and G. D. Durgin, "Complete link budgets for backscatterradio and RFID systems," *IEEE Antennas Propag. Mag.*, vol. 51, no. 2, pp. 11–25, Apr. 2009.
- [50] P. V. Nikitin and K. V. S. Rao, "Antennas and propagation in UHF RFID systems," in *Proc. IEEE Int. Conf. RFID*, Las Vegas, NV, USA, Apr. 2008, pp. 277–288.
- [51] G. Li, M. Chen, F. Zhao, Z. Li, D. Inserra, G. Li, and G. Wen, "A reflectionamplifier-based long-range semi-passive UHF RFID tag with standard compatibility," *IEEE Microw. Wireless Technol. Lett.*, vol. 33, no. 9, pp. 1373–1376, Sep. 2023.

HECTOR SOLAR received the M.Sc. degree in telecommunication engineering from the University of Pais Vasco, Bilbao, Spain, in 2002, and the Ph.D. degree in electronic engineering from the University of Navarra (Tecnun), Technological Campus, San Sebastian, Spain, in 2007.

From 2007, he was a Researcher with the Electronics and Communications Division, CEIT, San Sebastian, Spain. Through CEIT, he has collaborated in the design RF/mm-wave, low-power sensors, and RFID devices in CMOS. From 2006 to 2007, he was an External Consultant with Seiko–Epson, Barcelona, Spain. He is currently an Associate Professor with the Department of Electrical and Electronic Engineering, Tecnun. He has authored or coauthored more than 40 international journals and conference papers. He holds one patent. He coauthored two books *Linear CMOS RF Power Amplifiers* (Springer, 2014) and *Systems Design for Remote Healthcare* (Springer, 2014). His research interests include CMOS RF/mm-wave integrated circuit design and low power analog circuit design with a special emphasis on battery-less sensor nodes.

ANDONI BERIAIN (Member, IEEE) received the M.Sc. and Ph.D. degrees in telecommunication engineering from the University of Navarra (Tecnun), San Sebastian, in 2008 and 2013, respectively, with a focus on low-power sensor interface design.

From 2013 to 2018, he was devoted to the design of the analog front-end and mixed-signal blocks of a sensor-enabled passive RFID chip marketed as Rocky100 by Farsens Company. He is currently an Associate Professor with the Department of Electrical and Electronic Engineering, Tecnun. His current research interests include long-range RFID sensor applications, and energy harvesters and power management units for autonomous sensor nodes. **ROC BERENGUER** (Senior Member, IEEE) received the M.Sc. and Ph.D. degrees from the University of Navarra (Tecnun), Technological Campus, San Sebastian, Spain, in 1996 and 2000, respectively.

From 1999 to 2015, he was with CEIT, San Sebastian. Through CEIT and INCIDE (spin-off of the CEIT's COMMIC Group), he was an External Consultant with Siemens, Munich, Germany, in 2000; Hitachi Micro Systems Europe, Maidenhead, U.K., 2001; Xignal Technologies, Munich, from 2001 to 2002; Seiko-Epson, Barcelona, Spain, from 2006 to 2007; and Innophase, San Diego, CA, USA, from 2012 to 2023, where he collaborated in the design of several RF front ends for wireless standards, such as GSM-EDGE, DAB, and Wibree. He is currently an Associate Professor with the Department of Electrical and Electronic Engineering, Tecnun. He is the author or coauthor of more than 70 refereed publications in journals and conferences. He holds ten patents. He is the coauthor of the books Design and Test of High Quality Integrated Inductors for RF Applications in Conventional Technologies (Springer, 2003), GPS and Galileo: Dual RF Front-End Receiver Design, Fabrication, and Test (McGraw-Hill, 2008), and Linear CMOS RF Power Amplifiers: A Complete Design Workflow (Springer, 2014). His technical expertise and research interests include CMOS RF/mmwave IC design, ultralow-power analog circuit design for battery-less sensor nodes, and high-speed signal processing.

Dr. Berenguer is an Assessor of the Spanish Agency of Evaluation and Prospective (ANEP). He served as a TPC member for the IEEE European Solid State Circuit Conference, IEEE Midwest Symposium Circuits and Systems, and the IEEE Ph.D. Research in Microelectronics and Electronics. He also served as a Reviewer for several journals, such as IEEE JOURNAL OF SOLID-STATE CIRCUITS, IEEE TRANSACTIONS ON MICROWAVE THEORY AND TECHNIQUES, IEEE TRANSACTIONS ON CIRCUITS AND SYSTEMS— I: REGULAR PAPERS, and IEEE TRANSACTIONS ON CIRCUITS AND SYSTEMS—II: EXPRESS BRIEFS. **JAVIER SOSA** received the M.S. and Ph.D. degrees in telecommunication engineering from the University of Las Palmas de Gran Canaria, Spain, in 1999 and 2006, respectively.

Since 1999, he has been with the Institute for Applied Microelectronics (IUMA) working on CAD and VLSI design of emergent technologies, performance estimation and optimization of circuits and systems, and functional and formal verification. In 2000, he was with the Division of Design and Optimization of Very High-Speed Digital Communication Systems, Vitesse Semiconductor Corporation, Camarillo, CA, USA. From 2001 to 2008, he was a Lecturer with the Department of Electronic Engineering, University of Las Palmas de Gran Canaria, where he is currently an Associate Professor. In 2009, he was a Visiting Professor in VLSI multimedia signal processing with the Department of Information and Communications, Gwangju Institute of Science and Technology, Gwangju, South Korea. His current research interests include the design, performance estimation, and optimization of digital integrated circuits and systems oriented to sensors, and sensor networks, and its applications.

JUAN A. MONTIEL-NELSON (Senior Member, IEEE) received the M.S. degree in electrical engineering and the Ph.D. degree in electronics engineering, in 1991 and 1994, respectively.

From 1996 to 1997, he was a Visiting Scientist with the Centre for Very High-Speed Microelectronic Systems and the Department of Computer and Communication Engineering, Edith Cowan University, Australia. He is currently a Full Professor in instrumentation and microelectronics with the Institute for Applied Microelectronics, University of Las Palmas de Gran Canaria. He has authored or coauthored more than 150 papers in edited books, international journals, and conference proceedings. His current research interest includes the design and integration of biological and bio-medical sensor devices. He was a recipient of the Myril B. Reed Best Paper Award from the 2007 IEEE Midwest Symposium on Circuits and Systems.

...